Author's response to reviews

Title: Clinicians' caseload management behaviours as explanatory factors in patients' length of time on caseloads: a predictive multilevel study in paediatric community occupational therapy

Authors:

    Niina Kolehmainen (n.kolehmainen@abdn.ac.uk)
    Graeme MacLennan (g.maclennan@abdn.ac.uk)
    Jillian J Francis (j.francis@abdn.ac.uk)
    Edward AS Duncan (edward.duncan@stir.ac.uk)

Version: 7 Date: 20 August 2010

Author's response to reviews:

We have made a couple of very minor changed (related to commas and changing 'and' to 'or'). We have also added information to two of the researchers’ affiliations.